Author’s response to reviews

Title: Can Body Mass Index influence the fracture zone in the fifth metatarsal base? A retrospective review

Authors:
Mattia Pugliese (mattiapugliese@outlook.com)
Daniele De Meo (dannydemeo@hotmail.com)
Ennio Sinno (enniosinno@gmail.com)
Virginia Pambianco (v.pambianco@hotmail.com)
Armando Cavallo (armandocavallo90@gmail.com)
Pietro Persiani (pietro.persiani@uniroma1.it)
Ciro Villani (ciro.villani@uniroma1.it)

Version: 3 Date: 09 Jan 2020

Author’s response to reviews:

Please consider the title of your submission and whether your manuscript is able to determine relationships.

The title was changed from "The relationship between Body Mass Index and the localization of fifth metatarsal base fracture: a retrospective review" to "Can Body Mass Index influence the fracture zone in the fifth metatarsal base? A retrospective review"

Please upload a clean version of your submission so that the contents can be reviewed.

The manuscript was updated in a clean version